Author’s response to reviews

Title: Single-center study of Lynch syndrome screening in colorectal polyps

Authors:

fangchao zhu (zhufangchao85@163.com)
Da Pan (panda504@126.com)
Hui Zhang (51644879@qq.com)
Qiong Ye (1308368813@qq.com)
PeiSong Xu (550742305@qq.com)
JIE PAN (26996360@qq.com)

Version: 3 Date: 27 Feb 2019

Author’s response to reviews:

The grammatical errors have been revised by two English language experts. Just like other studies, also we didn’t set positive controls about IHC assays. We selected nuclear staining of non-tumor cells such as normal colonic epithelial cells, lymphoma cells, or stromal cells as an internal positive control. We judged that there was no nuclear staining at all in tumor cells and suggested that MMR was deficient.